Author's response to reviews

Title: An educational intervention to reduce pain and improve pain management for Malawian people living with HIV/AIDS and their family carers: study protocol for a randomised controlled trial

Authors:

Kennedy B Nkhoma (ntxkn1@nottingham.ac.uk)
Jane E Seymour (jane.seymour@nottingham.ac.uk)
Antony J Arthur (antony.arthur@uea.ac.uk)

Version: 2 Date: 29 June 2013

Author's response to reviews:

The document 'cover letter resubmission.doc' contains a detailed report of all revisions in response to reviewers comments